

# THE ROLE OF SUSTAINABLE FINANCE IN ACHIEVING SUSTAINABLE DEVELOPMENT GOALS: DOES IT WORK?

Magdalena ZIOLO<sup>1\*</sup>, Iwona BAK<sup>2</sup>, Katarzyna CHEBA<sup>3</sup>

<sup>1</sup>Faculty of Economics, Finance and Management, University of Szczecin, Szczecin, Poland <sup>2,3</sup>Faculty of Economics, West Pomeranian University of Technology, Szczecin, Poland

Received 02 February 2020; accepted 06 Septemeber 2020

**Abstract.** The World Bank and International Monetary Fund reported that the achievement of the 17 Sustainable Development Goals (SDGs) requires an escalation of development finance. The report Scaling Finance for the Sustainable Development Goals highlighted the urgency of the efforts to realize SDGs in encouraging financial innovation to move quickly. Even if the role of finance in achieving SDGs is unquestionable, few scientific studies have addressed these issues. We tried to fill the existing research gap. In this study, we examined the link between sustainable finance and SDGs based on European Union countries belonging to the OECD. We present a new and the original research approach. We assumed that the sustainable finance model plays a fundamental role in implementing SDGs (all SDGs were analysed except for SDG 6 and SDG14, due to lack of statistics were not analysed) and ensuring that social and environmental sustainability are reflected in SDGs. The results of this study show that the more sustainable the finance model, the better the achievement of SDGs in the group of analysed countries. We found a strong link between sustainable finance model and social sustainability (SDG1, 3, 4, 5, 10, 16); environmental sustainability (SDG11, 12, 13, 15) and economic sustainability (SDG8, 9, 17).

Keywords: sustainable development goals, SDGs, Agenda 2030, sustainable finance model, the correspondence analysis.

JEL Classification: G14, G15, G18, G21, G23, O16, O44, Q01.

## Introduction

The 2030 Agenda for Sustainable Development (Agenda 2030) contains 17 Sustainable Development Goals (SDGs) that are integrated and linked together (United Nations, 2015). This means that SDGs and the effectiveness of policies implemented to achieve each of the goals are also interrelated (Collste et al., 2017). SDGs generally refer to actions and policies that aim to mitigate the negative externalities of human activity and especially referring to the scope of social inclusiveness (SDGs 1–11, 15, 16), ecological inclusiveness (SDGs: 1, 2,

\*Corresponding author. E-mail: magdalena.ziolo@usz.edu.pl

Copyright © 2020 The Author(s). Published by Vilnius Gediminas Technical University

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons. org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited. 6–9, 11–15), and relational inclusiveness (SDGs: 3, 4, 10–14, 16, 17) (Gupta & Vegelin, 2016). Interactions occur among SDGs that can be both positive and negative, usually depending on key factors such as geographical context, resource endowments, time horizon, and governance (Nilsson et al., 2018).

To achieve the SDGs, financing is necessary. The International Monetary Fund [IMF] published an analysis that showed that developing countries face an average annual funding gap of some USD \$2.6 trillion for investment in health, education, roads, electricity, water, and sanitation. For low-income developing countries, this means additional annual spending that can amount to as much as 15% of their gross domestic product (GDP) (United Nations, 2019). The enormous and unmet need of developing countries may lead to the conclusion that the achievement of the SDGs requires an increased funding from billions to trillions (International Monetary Fund [IMF], 2015).

In 2015 in Addis Ababa, the third Financing for Development conference was held, and at the top of the issues discussed during the event, domestic resource mobilisation (DRM) and development was identified as an effective solution to and method of financing sustainable development (Long & Miller, 2017). Besides domestic resources, the role of financial markets in financing sustainable development is crucial and has been discussed in literature, especially in the context of capital market role (United Nations Global Compact, 2019).

A broad discussion exists in literature about the relationship between finance and sustainability (Aspinall et al., 2018; Ferreira et al., 2016). The overall conclusion based on the research results is that conventional finance is inadequate and unsuitable for financing SDGs as the three-dimensional perspective of sustainable development is not considered, leaving no room for environmental and social issues (Pisano et al., 2012; Fullwiler, 2015). Some postulates have been formulated in the scope of financing sustainable development that may improve financing, such as (1) incorporating non-financial (ESG) factors into risk analysis of financial institutions (OECD, 2018); (2) developing sustainable finance roadmaps and increasing cooperation regionally, and (3) developing typology of sustainable assets and finance (Zorlu, 2018). Sustainable finance is a common part of the proposed postulates.

Sustainable finance is a developing concept and new paradigm of finance. It is well defined in the literature. The general definition of sustainable finance explains this kind of finance as finance that takes into consideration ESG factors into financial decision-making (Schoenmaker, 2017). The subcategories of sustainable finance refer to major forms of environmental degradation, and inequalities that are part of definitions of sub-disciplines of sustainable finance (Ziolo et al., 2019).

In the relationship between sustainable finance and negative externalities (Ziolo et al., 2019) we aimed to examine the link between SDGs, and the finance model. First, relationships between sustainable finance, sustainable development and SDGs were analyzed in the context of negative externalities (Principles for Responsible Investment, 2017; Schoenmaker, 2017; United Nations, 2016; IMF, 2019; Ziolo et al., 2019). Then, a risk analysis was made in terms of the strength of its impact to justify which types of risk and the negative externalities that accompany them are the most relevant from the point of view of economy and finance and global challenges. For this purpose Global Risk Report 2020 was analyzed. At the top of the list of 10 risks in terms of impact according to Global Risk Report 2020 was Climate action failure. The group of environmental risk in top 10 is the most representatives by environmental risk (5 among 10) two risks are societal (social), two technological and one geopolitical. In sum, environmental risk is the most important one taking in terms of impact (World Economic Forum, 2020). This is also confirmed for leading global challenges defined as climate change, poverty, inequalities. The same is a truth for sustainable finance, according to IMF 2019 "Losses from climate- related risks affect the financial system directly, through price impairment, reduced collateral values, and underwriting losses, and indirectly, through lower economic growth and tighter financial conditions" (IMF, 2019). Based on the analysis of risks and challenges environmental damage is a crucial among negative externalities (Centemeri, 2009), it's also one of an effect of climate change (Kahn et al., 2019) and climate change is a major example of an integrated approach spanning a range of goals (Elder & Olsen, 2019). Finally we analysed availability and comparability of data set and data sources for period of time and countries. With the exception of SDG6 and SDG14, data set is comparable and available. SDG6 is represented by 6 variables (indicators), but none of them is available for all EU countries. In some case the missing data concern over 10 countries. The missing data is observed in the case of countries without access to the sea i.e. Czechia, Luxembourg, Hungary, Austria, and Slovakia. It is the main reason that according to this goal the EU countries could not to be compared.

This approach is original and novel. To the best of our knowledge, this is one of the first publications assuming a link between the sustainable finance model and achievement of SDGs. All publications have been published so far focused on funding sources, not models of finance. The original approach is therefore based on the assumption that the sustainable finance model determines the achievement of SDGs. To date, the literature on the subject has focused on analyzing only amounts necessary to finance SDGs or sources of funds (public/private), while the financing model has not been studied at all. An innovative approach also lies in the fact that the study includes sustainable finance, while previous studies have not made a distinction at all between conventional and sustainable finance based only on conventional financing.

The main research hypothesis is that the more sustainable the finance model, the better the link with achievement of SDGs. The specific research questions were defined as follows:

- (1) For which SDGs is the link to the sustainable finance model the strongest?
- (2) For what groups of countries are the strongest relationships observed between SDGs and the sustainable finance model?
- (3) What factors determine the differences between countries?

The remainder of this manuscript is organized as follows: Section II presents theoretical framework. Section III describes the data, variables assigned to describe the SDGs; Section IV explains research method. Section V provides the research results and discussion. Section VI outlines our conclusion and recommendations.

#### 1. Literature review

Scholtens (2006) emphasizes finance as a driver of sustainability, especially through socially responsible investments (Waring & Edwards, 2008). In the last decades many institutions point out the need for financial institutions to integrate environmental, social and corporate governance factors (ESG factors) into the decision-making process to mitigate ESG risk.

According to Pisano et al. (2012), a vast gap remains between sustainable development and the actions of most financial markets. Vandekerckhove and Leys (2012) identify especially issues that must be revised to cover the gap between sustainable development and finance among them: better indicators for analysing sustainable development goals (SDGs); recommendations for sustainable financing strategies and investments (Ziolo et al., 2019). Sustainable finance is developing concept and a kind of response to financial markets to sustainable development challenges related to its financing. Gerster (2011) points out that sustainable finance is defined as a kind of financing addressing environmental, social, and governance (ESG) impacts of financial services. Schoenmaker (2017) propose framework for Sustainable Finance based on sustainable finance models (SFM). Schoenmaker (2017) distinguishes SF 1.0 - Profit maximisation, while avoiding "sin" stocks; SF 2.0 - Internalisation of externalities to avoid risk; SF 3.0 - Contributing to sustainable development, while observing financial viability. Interdependencies between finance and sustainable development are the most commonly analysed in the context of: ESG risk and integrating non-financial factors into business practices (Nikolakis et al., 2012); ESG risk and financial performance (Edmans, 2011; Gompers et al., 2003); financial markets versus global warming and civil libertie (Alm & Sievänen, 2013); impact investment (Hebb, 2013) and socially responsible investment (Vandekerckhove & Leys, 2012). Ferreira et al. (2016) present a systematic review of literature about finance and sustainability in accordance to the thematic fields as follows: investors in general; SRI; governance over impact investment; institutional investors; climate change and human rights; non-renewable extractive industry; and sustainable development. Based on systematic literature review Ferreira et al. (2016) argue there is the research gap which should be filling by new research referring the scopes of sustainable finance. Waygood (2011) states that financial institutions may impact on corporate sustainability in two ways: via financial performance and investor advocacy influence. Jeucken (2004) declares financial institutions are often significant actor in a society's progress toward sustainable development. Besides social impact of finance on society also environmental impact is reported in literature review. Chen (2013) states green finance is the key of low carbon economy and the development of low carbon economy is not possible without the green finance. Interdependencies and impact of finance on society and environment is a scope of interest of research globally. Environment and society are also a research subject in the field related to SDGs.

Literature review related to SDG analyse SDG in two ways – separately based on selected SDGs or as a SDGs network. The most common discussed among SDGs is health and environment for example Buse and Hawkes (2015) focused on health in the sustainable development goals and emphasized that his will require a paradigm shift in global health. Nerini et al. (2018) concentrate on SDG7 (energy) and characterize synergies and trade-offs between efforts to achieve SDG7. Gain et al. (2016) discuss water security in global context analyzing SDG6. Many studies focused on analysing the achievement and financing of SDGs in selected economy sectors like health (Barroy et al., 2018) or education (Rambla & Langthaler, 2016) or analysing the results considering the geographical location, especially Africa (The Sustainable Development Goals Center for Africa, 2017).

The network approach shows that some thematic areas covered by the SDGs are well connected with one another, hence other scopes of the network have weaker connections with the rest of the network (Le Blanc, 2015). Nilsson et al. (2016) demonstrate a simple way of rating relationships between the SDG using Goals scoring. Hajer et al. (2015) proposed more general approach to SDGs based on four connected perspectives: "planetary boundaries", "the safe and just operating space", "the energetic society", "green competition". Gupta and Vegelin (2016) researched interactions between inclusive development and SDG and they argue there is a risk that theory (text about SDGs) and implementation processes focus more on social inclusiveness rather than on ecological and relational inclusiveness. Stafford-Smith et al. (2017) claim that there must be greater attention on interdependencies in three scopes: across sectors (e.g., finance, agriculture, energy, and transport), across societal actors (local authorities, government agencies, private sector, and civil society), and between and among low, medium and high income countries. Financial aspect related to SDGs is usually focused on investment or development finance context. Kedir et al. (2017) calculate the additional investment required to meet SDGs, with a focus on SDG 1. Schmidt-Traub and Sachs (2015) analyse private and public financing needs for the SDGs and formulate recommendations.

Publications on the relationship between finance and SDG focused on several threads, discussed public and private sources of SDG financing on a macro scale (Kharas et al., 2015), and the methods and instruments of financing selected SDGs on a micro scale (Gambetta et al., 2019). However, no researcher has examined the effectiveness and efficiency of SDGs financing depending on the financial model.

One of the few publications that identified the relationship between the SDG and fiscal policy instruments was Fiscal Policies and the SDGs in the Green Economy (UNEP, 2016). The report indicates fiscal instruments, such as taxes, fees, and public expenditure, and their role in stimulating and supporting the implementation of SDGs 6–14 and 17. The report raises the issue of the role of governments in shaping fiscal policy for sustainable development with sustainable fiscal policy instruments.

Sachs (2015) considered the financing mechanism for SDGs and discussed the financing of SDGs in the context of public (SDGs 3–7) and private mechanisms of financing (SDG13). Sachs raised the question of how we will ensure the required scale of flows and effective policies and institutional structures to manage the flows. Sachs emphasized that much of the necessary finance will flow through private markets, some will come from philanthropy and not-for-profit businesses, and much of it will need to come through the public sector, but no particular recommendations were provided about how to organize the systems, only general statements were outlined (Sachs, 2015).

The recommendations for financing sustainable development were considered in Financing the Sustainable Development Goals: Lessons from government spending on the SDGs. The report suggests doubling tax revenue, overhauling global tax rules, doubling concessional development cooperation, and improving the allocation and effectiveness of financing (Development Finance International & Oxfam, 2015). Kumar et al. (2016) recognized challenges related to achievement of SDGs, including: the huge cost of achieving the SDGs, the indicators for measuring SDG progress have not yet been identified, and a lack of accountability for inputs into SDGs at all levels. Klees (2017) stated that global taxation must replace the charity model.

# 2. Statistical data

To analyse the relationships between the area of sustainable development and finance, we used two data sets. The basis for analyses in the first area of this research was the indicators used by European Commission to monitor progress in the implementation of the latest strategy for sustainable development (Agenda 2030)<sup>1</sup>. In accordance with the assumptions presented in the introduction, we used the indicators describing the fifteen SDGs of this strategy in the second area of this research. Due to the availability of data, we analysed 2016 data. The following provides a detailed list of these indicators<sup>2</sup>:

- Goal 1: End poverty in all its forms everywhere (SDG1). SDG\_01\_10: *D*; SDG\_01\_20: *D*; SDG\_01\_30: *D*; SDG\_01\_40: *D*; SDG\_01\_50: *D*; SDG\_01\_60: *D*;
- Goal 2: End hunger, achieve food security and improved nutrition and promote sustainable agriculture (SDG2). SDG\_02\_10: *D*; SDG\_02\_20: (source: EC services), *S*; SDG\_02\_30: *S*; SDG\_02\_40: *S*; SDG\_02\_60: *D* (source: EEA);
- Goal 3: Ensure healthy lives and promote well-being for all at all ages (SDG3). SDG\_03\_10: S; SDG\_03\_20: S; SDG\_03\_40: D; SDG\_03\_50: D; SDG\_03\_60: D;
- Goal 4: Ensure inclusive and equitable quality education and promote lifelong learning opportunities for all (SDG4). SDG\_04\_10: D; SDG\_04\_10A: D; SDG\_04\_20: S; SDG\_04\_30: S; SDG\_04\_40: D, (source: OECD); SDG\_04\_50: S; SDG\_04\_60: S;
- Goal 5: Achieve gender equality and empower all women and girls (SDG5).
   SDG\_05\_20: D; SDG\_05\_30: D; SDG\_05\_40: D; SDG\_05\_50A: S; SDG\_05\_50B: S;
   SDG\_05\_60A: S, (source: EIGE); SDG\_05\_60B: S, (source: EIGE);
- Goal 7: Ensure access to affordable, reliable, sustainable and modern energy for all (SDG7). SDG\_07\_10: *D*; SDG\_07\_11: *D*; SDG\_07\_20: *D*; SDG\_07\_30: *S*; SDG\_07\_40: *S*; SDG\_07\_50: *D*; SDG\_07\_60: *D*;
- Goal 8: Promote sustained, inclusive and sustainable economic growth, full and productive employment and decent work for all (SDG8). SDG\_08\_10: *S*; SDG\_08\_11: *S*; SDG\_08\_20: *D*; SDG\_08\_20A: *D*; SDG\_08\_30: *S*; SDG\_08\_30A: *S*; SDG\_08\_40: *D*; SDG\_08\_60: *D*;
- Goal 9: Build resilient infrastructure, promote inclusive and sustainable industrialization, and foster innovation (SDG9). SDG\_09\_10: *S*; SDG\_09\_20: *S*; SDG\_09\_30: *S*; SDG\_09\_40: *S*; SDG\_09\_60: *S*;
- Goal 10: Reduce inequality within and among countries (SDG10) SDG\_10\_10: S; SDG\_10\_20: S; SDG\_10\_30: D; SDG\_10\_50: D;
- Goal 11: Make cities and human settlements inclusive, safe, resilient and sustainable (SDG11) SDG\_11\_10: *D*; SDG\_11\_20: *D*; SDG\_11\_31: *S*; SDG\_11\_40: *D*; SDG\_11\_60: *S* (source: EC services);
- Goal 12: Ensure sustainable consumption and production patterns (SDG12) SDG\_12\_20: S; SDG\_12\_30: D; SDG\_12\_41: S; SDG\_12\_50: D;
- Goal 13: Take urgent action to combat climate change and its impacts (SDG13). SDG\_13\_10, *D*; SDG\_13\_20: *D*; SDG\_13\_50: *S*;

<sup>&</sup>lt;sup>1</sup> All indicatoros were retrieved from https://ec.europa.eu/eurostat/data/database, the codes of indicators according to https://ec.europa.eu/eurostat/data/database.

<sup>&</sup>lt;sup>2</sup> S and D describe the character of indicator: S is dedicated for stimulants, and D indicates for destimulants.

- Goal 15: Protect, restore and promote sustainable use of terrestrial ecosystems, sustainably manage forests, combat desertification, and halt and reverse land degradation and halt biodiversity loss (SDG15). SDG\_15\_10: S; SDG\_15\_20: S (source: DG ENV, EEA); SDG\_15\_41: D (source: EEA); SDG\_15\_50: D (source: JRC);
- Goal 16: Promote peaceful and inclusive societies for sustainable development, provide access to justice for all and build effective, accountable and inclusive institutions at all levels (SDG16). SDG\_16\_10 *D*; SDG\_16\_20: *D*; SDG\_16\_30: *S*; SDG\_16\_40: *S* (source: DG COMM); SDG\_16\_50: *S*; SDG\_16\_61: *S*; SDG\_16\_62: *S*; SDG\_16\_63: *S* (source: EC services), European Central Bank (ECB);
- Goal 17: Strengthen the means of implementation and revitalize the global partnership for sustainable development (SDG17). SDG\_17\_10: *S* (source: OECD), *S*; SDG\_17\_30: *S*; SDG\_17\_40: *D*; SDG\_17\_50A: *S*; SDG\_17\_50B: *S*.

Referring to sustainable finance (SF) models, we based our approach on that of Schoenmaker (2017). Schoenmaker distinguished finance-as-usual (F); SF 1.0: profit maximisation, while avoiding "sin" stocks (F > S and E); SF 2.0: internalisation of externalities to avoid risk (T = F + S + E); and SF 3.0: contributing to sustainable development while observing financial viability (S and E > F)<sup>3</sup>.

In the area containing indicators describing the sustainable finance (F) models in the analysed EU countries, the following indicators were considered: F1 (government support to agricultural research and development, Euro per inhabitant, S), F2 (Gini coefficient of equivalised disposable income, coefficient of 0 (maximal equality) to 100 (maximal inequality, D), F3 (government expenditure on "education" as a ratio to GDP, % of GDP, S), F4 (government expenditure on health as a ratio to GDP, % S), F5 (government expenditure on social protection, as a ratio to GDP, % S), F6 (government expenditure on law courts, Euros per inhabitant, S), F7 (government gross debt, as a ratio to GDP, %, S), F8 (shares of labour taxes in total tax revenues, percent of total taxes, S), F9 (share of environmental tax revenue in total government revenue from taxes, %, S), F10 (percent of total revenues from environmental taxes and social contributions, S), F11 (consolidated banking leverage, domestic and foreign entities (asset-to-equity multiple), D), F12 (bank credit to the private sector as percent of GDP, S), F13 (official development assistance as share of gross national income, percent of gross national income (GNI), S), F14 (European Union (EU) imports from developing countries, in million euros per 100,000 inhabitants, S), and F15 (income from natural resources, percent of GDP, S).

The indicators explaining the sustainable finance scope were selected to reflect the threedimensional nature of sustainable finance, where variables: F1, F8, F9, F10, and F15 are related to the environmental sustainability and sustainable finance; variables: F2, F3, F4, F5, and F6 are related to the social pillar; and variables: F7, F11, F12, F13, and F14 are related to then economic pillar. When selecting the indicators within the sustainable finance scope, we based our approach on that presented by: Apergis et al. (2013) (environmental pillar), Burchardt and Vizard (2007) (social pillar) and economic (Čihák et al., 2012). The reference indicator framework to monitor the SDGs in an EU context was established by European

<sup>&</sup>lt;sup>3</sup> F – financial value; S – social impact; E – environmental impact; T – total value.

Commission in 2017. The variables set consist of at most 6 indicators per SDGs in order to guarantee the same importance to all goals (EU SDG indicator set 2020). The indicators related to: economic, social and environmental sustainability are crucial to evaluate the Agenda 2030 objectives.

Indicators that refer to environmental pillar of sustainable development (included in SDG2, SDG7, SDG9, SDG11, SDG12, SDG13, SDG15, SDG17) and sustainable finance (F1, F8, F9, F10, and F15 mitigate negative impact of externalities) are related to negative externalities – environmental degradation (SDG2, SDG11, SDG12, SDG13, SDG15 represent negative externalities describe as *D*; SDG13 one exception "commitment on climate-related expenditure" positive impact describe as S; variables in SDG7, SDG9, SDG17 mitigate negative externalities described as *D*). One of the crucial air pollution sources is the emission of CO2. CO2 is responsible for climate change (global warming). CO2 emissions from the agricultural production represent 21–25 percent of total CO2 emissions (FAO, 2020). The Authors like Oates (1995), Parry and Small (2005), Lin and Li (2011) argue that environmental taxes are responsible for reducing the impact of pollution on environmental devastation. The correlation between public spending on R&D and GHG is also observed (Apergis et al., 2013; Lee & Min, 2015).

The group of financial variables like: F2, F3, F4, F5, and F6 refer to social pillar of sustainable finance. The variables related to SDG1, SDG2, SDG3, SDG4, SDG5, SDG10, SDG11, SDG16 described *D* indicate negative externality (social exclusion) and described S represent positive impact, mitigate it. We used Burchardt and Vizard (2007) research results to explain the selection of variables in the scope of: life (variables: SDG1, SDG2, SDG3, SDG4 SDG10), individual family and social life (variables: SDG5, SDG10), health (variables SDG3, SDG4 SDG16), standard of living (variables SDG1, SDG11, SDG16); taking into account their coherence with negative externalities related to social pillar of sustainable development and sustainable finance and the way one can mitigate it. Taking into account the framework of social exclusion in European Union, social exclusion is determined by labour market position (SDG1, SDG5, SDG9, SDG10); life history (SDG1); social class; capital/welfare resources (SDG1, SDG4, SDG5); health wellbeing (SDG3, SDG16) (European Communities, 2002, p. 21).

The variables related to economic pillar of sustainability and to sustainable finance include indicators: F7, F11, F12, F13, and F14. The variables F11, F12 refer to financial soundness (Čihák et al., 2012). The indicators F7, F13, F14 correspond with economic development and welfare. According to Rubin (2011) the development and adoption of new technology is a comprehensive response to global climate change. Rubin (2011) points out that R&D is a critical element of the policy portfolio needed to foster innovations that reduce GHG emissions. Reducing GHG emissions is possible through innovations in technology and institutions will require increased numbers of skilled workers. This explains the economic variables included in SDG8, SDG9, SDG10, SDG17. The role of inclusive innovations in inclusive growth and social inclusion is crucial one and coherent with SDG9 (Heeks et al., 2013). It is worth mentioning, the selection of indicators was determined by Eurostat data set availability and comparability for period of time and countries.

### 3. Research method

To examine the relationships between the indicated areas, a two-stage research procedure was used. In the first stage, to calculate rankings in each of the distinguished areas, we separately considered a taxonomic measure of development based on standardized sums. The application of a taxonomic measure allowed us to compare the development of objects (in this case, EU countries belonging to the OECD). The higher the value taken by the taxonomic measure, the higher the level of the phenomenon marked by the object is. The construction of taxonomic measure uses different methods to normalize diagnostic indicators. In the study, we standardized the indicators. In this method, every diagnostic indicator is considered equally important. Based on the standardized sum method, the synthetic measure was calculated as follows (Bąk et al., 2019):

$$g_{i} = \frac{1}{m} \left[ \sum_{j \in S} \frac{y_{ij} - \bar{y}_{j}}{s_{j}} + \sum_{j \in D} \frac{\bar{y}_{j} - y_{ij}}{s_{j}} \right],$$
(1)

where: S – set of indicators that are stimulants; D – set of destimulants;  $\overline{y}_j$  – mean value;  $s_j$  – standard deviation.

In the next step to the calculation of taxonomic measure of development the following formula can be applied:  $g'_i = g_i - \min_i g_i$ ,  $g''_i = \frac{g'_i}{\max_i g'_i}$ , i = 1, ..., The application of the formula can be applied:

mula for  $g_i$ ' shifts the scale of the  $g_i$  measure to the zero point, and the transformation of  $g_i$ '' leads to the upper limit of the taxonomic measure equal 1. Finally, taxonomic measures are in the range <0, 1>. Based on the value of the taxonomic measure, four typological groups of EU countries can be distinguished: group A:  $g_i^{"} \ge \overline{g_i^{"}} + s_{g_i^{"}}$ , group B:  $\overline{g_i^{"}} + s_{g_i^{"}} \ge \overline{g_i^{"}} > \overline{g_i^{"}} > \overline{g_i^{"}}$ , group C:  $\overline{g_i^{"}} \ge \overline{g_i^{"}} - s_{g_i^{"}}$ , group D:  $\overline{g_i^{"}} - s_{g_i^{"}}$ .

In the second stage, to deeply analyse the relationships between considered areas, we used correspondence analysis. This method was selected to detect relationships between the categories of synthetic measures calculated on the basis of the value of diagnostic indicators belonging to the selected five SDGs and sustainable finance. A detailed description of this method can be found in many reports (Greenacre, 1984, 1993; Goodman, 1986; Clausen, 1998; Bak et al., 2019). Here, we applied the modified version of this method based on the values of taxonomic measure of development, transformed on a nominal scale. The results of division of EU countries into four typological groups were the basis of analyses presented in the paper. This means that in the modified version of correspondence analysis, the six synthetic measures obtained as a result of the division of EU countries in each analysed area that were divided into typological groups were the basis of calculation. For each of these variables, four categories were obtained, of which the first was associated with the highest values of the synthetic measure, and the fourth with the lowest. Finally, the categories of individual measure were presented as follows: taxonomic measure for Goal 1, SDG1:1 (best situation); SDG1:2, SDG1:3, and SDG1:4 (worst situation). The same method was applied to the rest of the considered variables representing the results achieved by EU countries for Goals 3, 9, 10, and 13, as well as in the area of sustainable finance, where F1 indicates countries with the best situation in this area and F4 indicates the worst.

In the first step of correspondence analysis, a complex contingency table (Burt matrix, contingent, cross-tabulation) included the number of all categories of measure, was applied to describe *n* objects (Greenacre, 1984, p. 140; 1994, p. 141; Andersen, 1991, p. 387; Lebart et al., 1984, p. 84). This is a matrix of *Z* variables that includes blocks (submatrices) referring to consecutive measures:  $\mathbf{Z} = [\mathbf{Z}_1, ..., \mathbf{Z}_Q]$ , where **Q** is the number of characteristics. The Burt matrix is calculated on the basis of an equation:  $\mathbf{B} = \mathbf{Z}^T \mathbf{Z}$ , which produces the symmetrical block matrix with diagonal matrices. In these matrices the number of categories of characteristics is contained. While the contingency tables for each pair of analysed variables outside the diagonal matrix are calculated. The dimension of actual space of the coexistence of category of variables is calculated as:

$$K = \sum_{q=1}^{Q} (J_q - 1) .$$
 (2)

The procedure requires to choose the best size of projection (dimensional space) of the categories of variables. For these purpose the Greenacre's criterion (Greenacre, 1984, p. 145; 1993, p. 145; 1994, p. 156) according to which the eigenvalues should be calculated taking into account the following condition:

$$\lambda_{B,k} > \frac{1}{Q}.$$
(3)

In the next step, the improvement of the results of the correspondence analysis presented in form of Burt matrix should be provided. The following formula is applied for this purpose:

$$\tilde{\lambda}_{k} = \left(\frac{Q}{q-1}\right)^{2} \cdot \left(\sqrt{\lambda_{B,k}} - \frac{1}{Q}\right)^{2}, \qquad (4)$$

where: Q – number of variables;  $\lambda_{B,k} - k$ -th eigenvalue.

New (modified) coordinate values in selected dimensional space for particular categories of variables were calculated as follows (Stanimir, 2005):

$$\tilde{\mathbf{F}} = \mathbf{F}^* \cdot \boldsymbol{\Gamma}^{-1} \cdot \tilde{\boldsymbol{\Lambda}},\tag{5}$$

where:  $\tilde{\mathbf{F}}$  – matrix (*n*×*m* dimensions) of new coordinate values for the variable categories;  $\mathbf{F}^*$  – matrix (with dimensions of *n*×*m*) of the original coordinate values for the variable categories;  $\tilde{\mathbf{A}}^{-1}$  – diagonal inverse matrix of singular values;  $\tilde{\mathbf{A}}$  – diagonal matrix of modified eigenvalues (dimension).

The final result of correspondence analysis is the presentation of the simultaneous occurrence of the categories of variables in a graphic form (Greenacre & Hastie, 1987, pp. 437–447; Goodman, 1986, pp. 243–309). In this case, it is necessary to decide the number of dimensions that allow the best mapping of the results obtained. It is worth paying attention that if a space larger than three is, usually to present the coexistence of characteristics the methods of classification can be applied (Bąk, 2013, p. 135). The values of projection coordinates of each category are defined as variables, and the categories of all analysed characteristics are objects. The methods of classification are usually applied when the number of all characteristics and its options is high. In this case the dispersion of points in the graph is an obstacle to distinguishing the classes unambiguously.

## 4. Results and discussion

The detailed results of the first stage of procedure described in the previous paragraph are presented in Table 1. First of all, consideration needs to be given to the seven states that were in the first three places in the rankings and in the first typological group, at least in the case of three goals. These include: Sweden (goals: 1, 4, 5, 9, 13, 15 and 17), the Netherlands (goals: 3, 8, 11, 12, 17), Germany (goals: 8, 9, 10, 13), Luxembourg (goals: 3, 7, 10, 16), Denmark (goals: 5, 9, 16), Finland (goals: 1, 15, 16) and France (goals: 3, 12, 13). Among them are three countries (Denmark, the Netherlands and Sweden), which are also ahead in the field of sustainable financing. Particularly noteworthy is Sweden, which found itself in the first typological group due to sustainable finance and in the case of eight goals (1, 4, 5, 8, 9, 13 15, 16, 17). For four purposes (2, 3, 7 and 10) its position was also above the average in the studied group of countries, which allowed it to be included in the second typological group. Only in the case of objectives 11 and 12 this country was included in the third group.

Country	Country F			SDG1			SDG2		SDG3			SDG4			SDG5			
Country	1	2	3	1	2	3	1	2	3	1	2	3	1	2	3	1	2	3
Austria	0.520	7	В	0.642	6	В	0.807	4	Α	0.799	5	В	0.653	9	В	0.221	3	С
Belgium	0.492	8	В	0.326	16	В	0.126	22	D	0.634	11	В	0.515	14	С	0.583	2	В
Czechia	0.230	16	С	0.918	2	С	0.540	11	В	0.466	16	С	0.495	15	С	0.024	4	D
Denmark	1.000	1	Α	0.545	11	Α	0.466	12	С	0.677	10	В	0.748	5	В	0.775	1	Α
Estonia	0.610	6	В	0.575	9	В	0.764	6	В	0.397	20	С	0.647	10	В	0.340	3	С
Finland	0.705	4	Α	1.000	1	Α	0.291	17	С	0.633	12	В	0.631	11	В	0.663	2	В
France	0.368	12	С	0.657	5	С	0.723	8	В	0.439	18	С	0.552	13	В	0.709	1	A
Germany	0.459	9	В	0.399	14	В	0.756	7	В	0.604	14	В	0.443	16	С	0.471	2	В
Greece	0.175	19	С	0.000	23	С	0.296	16	С	0.629	13	В	0.163	21	D	0.169	4	D
Hungary	0.105	22	D	0.349	15	D	0.446	14	С	0.466	17	С	0.223	20	D	0.000	4	D
Ireland	0.284	14	С	0.415	13	С	0.227	19	D	0.728	9	В	1.000	1	A	0.175	4	D
Italy	0.269	15	С	0.157	22	С	1.000	1	Α	0.761	7	В	0.000	23	D	0.530	2	В
Latvia	0.327	13	С	0.321	18	С	0.863	3	Α	0.059	22	D	0.423	17	С	0.388	3	С
Lithuania	0.000	23	D	0.223	20	D	0.593	10	В	0.000	23	D	0.684	8	В	0.630	2	В
Luxembourg	0.401	11	В	0.599	8	В	0.227	20	D	0.835	3	В	0.754	4	В	0.416	3	С
Netherlands	0.784	2	Α	0.572	10	Α	0.194	21	D	0.915	2	А	0.710	7	В	0.546	2	В
Poland	0.186	18	С	0.530	12	С	0.252	18	D	0.320	21	D	0.720	6	В	0.309	3	С
Portugal	0.163	20	С	0.299	19	С	0.431	15	С	0.593	15	С	0.265	19	С	0.343	3	С
Slovakia	0.199	17	С	0.752	4	С	0.889	2	Α	0.417	19	С	0.035	22	D	0.149	4	D
Slovenia	0.639	5	В	0.600	7	В	0.000	23	D	0.788	6	В	0.827	3	A	0.604	2	В
Spain	0.140	21	D	0.172	21	D	0.685	9	В	1.000	1	A	0.300	18	С	0.388	3	С
Sweden	0.729	3	Α	0.862	3	Α	0.775	5	В	0.814	4	В	0.859	2	A	1.000	1	A
United Kingdom	0.425	10	В	0.325	17	В	0.466	13	С	0.747	8	В	0.608	12	В	0.351	3	С

Table 1. The results of rankings in each considered areas of sustainable development and finance

	SD	G7		SI	G8		SE	G9		SD	G10		SD	G11		SD	G12	
Country	1	2	3	1	2	3	1	2	3	1	2	3	1	2	3	1	2	3
Austria	0.389	10	В	0.688	7	В	0.635	7	В	0.325	4	В	0.484	13	С	0.456	12	С
Belgium	0.000	23	D	0.439	15	С	0.625	8	В	0.281	7	В	0.658	5	В	0.546	8	В
Czechia	0.164	18	С	0.702	6	В	0.351	12	С	0.103	14	С	0.367	17	С	0.440	13	С
Denmark	0.610	4	Α	0.757	4	В	0.692	3	Α	0.273	8	В	0.506	11	В	0.611	5	В
Estonia	1.000	1	Α	0.639	9	В	0.241	15	С	0.055	19	С	0.509	10	В	0.000	23	D
Finland	0.325	11	С	0.629	10	В	0.650	5	В	0.243	11	В	0.400	14	С	0.230	22	D
France	0.119	20	D	0.403	17	С	0.648	6	В	0.251	10	В	0.378	15	С	0.816	2	Α
Germany	0.172	17	С	0.805	3	Α	1.000	1	Α	0.333	3	В	0.699	4	В	0.422	14	С
Greece	0.211	14	С	0.000	23	D	0.000	23	D	0.025	21	С	0.033	22	D	0.546	9	В
Hungary	0.118	21	D	0.449	14	С	0.258	14	C	0.013	22	С	0.338	18	С	0.419	15	С
Ireland	0.539	5	В	0.535	11	В	0.301	13	С	0.309	5	В	0.987	2	Α	0.489	10	В
Italy	0.159	19	С	0.110	22	D	0.227	17	С	0.187	12	С	0.569	8	В	0.749	3	Α
Latvia	0.653	3	Α	0.420	16	С	0.206	18	С	0.000	23	D	0.000	23	D	0.407	17	С
Lithuania	0.460	9	В	0.459	12	С	0.232	16	С	0.079	17	С	0.293	19	С	0.328	19	С
Luxembourg	0.673	2	Α	0.733	5	В	0.455	11	В	0.642	2	Α	0.554	9	В	0.419	16	С
Netherlands	0.181	16	С	0.923	2	Α	0.653	4	В	0.288	6	В	0.855	3	Α	1.000	1	Α
Poland	0.103	22	D	0.371	18	С	0.040	22	D	0.038	20	С	0.281	20	С	0.234	21	D
Portugal	0.513	7	В	0.358	19	С	0.122	20	D	0.091	16	С	0.198	21	D	0.608	6	В
Slovakia	0.313	12	С	0.318	20	С	0.178	19	С	0.064	18	С	0.374	16	С	0.323	20	С
Slovenia	0.515	6	В	0.456	13	С	0.459	10	В	0.091	15	С	0.616	6	В	0.463	11	С
Spain	0.282	13	С	0.222	21	D	0.094	21	D	0.139	13	С	0.595	7	В	0.603	7	В
Sweden	0.501	8	В	1.000	1	Α	0.851	2	Α	0.271	9	В	0.484	12	С	0.338	18	С
United Kingdom	0.189	15	С	0.663	8	В	0.489	9	В	1.000	1	A	1.000	1	A	0.706	4	A
		SDC	613				SDG15		I		SDO	G16				SDG17		
Country	1	2		3	1		2	3		1	2	2	3	1		2		3
Austria	0.429	14		C	0.50	)7	11	B		0.589	9		В	0.2		11		5
Belgium	0.374	15	;	С	0.26	59	17	С	;	0.628	8	3	В	0.1	63	14	(	2
Czechia	0.369	16	5	С	0.38	37	16	C	;	0.255	2	1	D	0.1	85	12	(	5
Denmark	0.621	5		В	0.73	37	4	В		0.822	3	3	А	0.5	36	4	1	ł
Estonia	0.042	22	2	D	1.00	00	1	А	L	0.460	1	2	С	0.7	70	2	1	ł
Finland	0.531	12	2	В	0.90	)3	3	А		0.927	2	2	А	0.2	67	9	(	2
France	0.816	2		Α	0.45	59	13	C	;	0.299	1	8	С	0.1	55	15	(	2
Germany	0.786	3		Α	0.59	95	8	В		0.660	7	7	В	0.3	74	6	I	3
Greece	0.522	13	3	В	0.56	55	10	В		0.000	2	3	D	0.1	16	18	(	0
Hungary	0.742	4		Α	0.09	91	20	D	)	0.437	1	4	С	0.0	55	20	(	5
Ireland	0.220	19	)	С	0.20	)8	18	C	2	0.689	6	5	В	0.1	16	19	(	2
Italy	0.556	10	)	В	0.6	56	6	В		0.231	2	2	D	0.2	53	10	(	2
Latvia	0.613	6		В	0.59	91	9	В		0.366	1	6	С	0.3	68	7	1	3
Lithuania	0.171	20	)	D	0.43	39	14	C	;	0.512	1	0	В	0.1	46	16	(	2
Luxembourg	0.000	23	3	D	0.00	51	21	D	)	1.000	1	L	А	0.3	84	5	1	3
Netherlands	0.103	21		D	0.00	00	23	D	)	0.733	5	5	В	1.0	00	1	1	4
Poland	0.233	18	3	С	0.0	55	22	D	)	0.446	1	3	С	0.0	39	21	(	2
Portugal	0.584	8		В	0.42	24	15	C	2	0.406	1	5	С	0.0	00	23	Ι	)

#### End of Table 1

Country	SDG13				SDG15			SDG16		SDG17			
Country	1	2	3	1	2	3	1	2	3	1	2	3	
Slovakia	0.553	11	В	0.197	19	D	0.347	17	С	0.171	13	С	
Slovenia	0.359	17	С	0.605	7	В	0.265	20	С	0.347	8	В	
Spain	0.576	9	В	0.677	5	В	0.267	19	С	0.030	22	D	
Sweden	1.000	1	Α	0.952	2	Α	0.796	4	Α	0.605	3	Α	
United Kingdom	0.592	7	В	0.490	12	В	0.492	11	С	0.129	17	С	

Note: 1 - means value of taxonomic measure of development, 2 - rank, 3 - typological group.

In all situations, the following countries were ranked the highest (in the first typological group): (1) implementation of Goal 1: Denmark, the Netherlands, and Sweden; (2) Goal 2 Austria, Italy, Latvia (3) Goal 3: the Netherlands and Spain; (4) Goal 4 Ireland, Sweden (5) Goal 5 Denmark, France, Sweden (6) Goal 7 Estonia, Denmark, Latvia, Luxemburg (7) Goal 8 Germany, the Netherlands, Sweden (8) Goal 9: Denmark, Germany, and Sweden; (9) Goal 10: Luxembourg and the United Kingdom; (10) Goal 11 Ireland, the Netherlands, the United Kingdom (11) Goal 12 France, Italy, the Netherlands, the United Kingdom (12) Goal 13: France, Germany, and Sweden; (13) Goal 15 Estonia, Finland, Sweden (14) Goal 16 Denmark, Finland, Luxemburg, Sweden (15) Goal 17 Denmark, Estonia, the Netherlands, Sweden (16) in the area of sustainable finance: Denmark, Finland, the Netherlands, and Sweden. Attention should be paid to Sweden, which was placed in the first typological group due to sustainable finance and achievement of the 1st, 4<sup>th</sup>, 5<sup>th</sup>, 8<sup>th</sup>, 9<sup>th</sup>, 13<sup>th</sup>, 15<sup>th</sup>, 16<sup>th</sup> and 17th goals. In the case of the remaining objectives (2, 3, 7 and 10), its position was also above average in the examined group of countries, which allowed it to be classified into the second typological group. The Netherlands and Denmark were also in a very good position, as it always placed high in the achievement of goals: 1, 3, 8, 11, 12, 13, 17 (Netherlands) and goals: 1, 5, 7, 9, 16, 17. For other countries, the situation was not so straightforward.

Table 2. Pearson correlation	coefficient in	the scope of	sustainable	finance and	the selected	goals of
Agenda 2030 in 2016		-				-

Goals:	SDG 1	SDG 2	SDG 3	SDG 4	SDG 5	SDG 7	SDG 8	SDG 9	SDG 10	SDG 11	SDG 12	SDG 13	SDG 15	SDG 16	SDG 17	F
SDG1	1.00	-0.02	0.06	0.41	0.20	0.13	0.62	0.47	0.07	0.07	-0.30	0.01	0.12	0.45	0.34	0.49
SDG2	-0.02	1.00	-0.25	-0.45	0.01	0.11	-0.08	0.00	-0.14	-0.24	-0.11	0.44	0.41	-0.24	0.03	-0.17
SDG3	0.06	-0.25	1.00	0.13	0.15	-0.12	0.22	0.33	0.46	0.61	0.45	0.05	0.05	0.25	0.20	0.42
SDG4	0.41	-0.45	0.13	1.00	0.39	0.37	0.66	0.44	0.33	0.42	-0.15	-0.34	0.03	0.61	0.38	0.52
SDG5	0.20	0.01	0.15	0.39	1.00	0.14	0.35	0.58	0.17	0.13	0.15	0.25	0.47	0.42	0.42	0.60
SDG7	0.13	0.11	-0.12	0.37	0.14	1.00	0.25	-0.09	-0.07	-0.09	-0.47	-0.31	0.41	0.26	0.41	0.28
SDG8	0.62	-0.08	0.22	0.66	0.35	0.25	1.00	0.76	0.44	0.44	-0.07	-0.04	0.08	0.75	0.65	0.66
SDG9	0.47	0.00	0.33	0.44	0.58	-0.09	0.76	1.00	0.46	0.45	0.17	0.31	0.22	0.63	0.47	0.71
SDG10	0.07	-0.14	0.46	0.33	0.17	-0.07	0.44	0.46	1.00	0.67	0.32	-0.03	-0.08	0.48	0.08	0.30
SDG11	0.07	-0.24	0.61	0.42	0.13	-0.09	0.44	0.45	0.67	1.00	0.33	-0.20	-0.12	0.40	0.26	0.37
SDG12	-0.30	-0.11	0.45	-0.15	0.15	-0.47	-0.07	0.17	0.32	0.33	1.00	0.18	-0.30	-0.13	0.05	0.09
SDG13	0.01	0.44	0.05	-0.34	0.25	-0.31	-0.04	0.31	-0.03	-0.20	0.18	1.00	0.38	-0.16	-0.23	0.04
SDG15	0.12	0.41	0.05	0.03	0.47	0.41	0.08	0.22	-0.08	-0.12	-0.30	0.38	1.00	-0.03	0.23	0.39
SDG16	0.45	-0.24	0.25	0.61	0.42	0.26	0.75	0.63	0.48	0.40	-0.13	-0.16	-0.03	1.00	0.43	0.58
SDG17	0.34	0.03	0.20	0.38	0.42	0.41	0.65	0.47	0.08	0.26	0.05	-0.23	0.23	0.43	1.00	0.75
F	0.49	-0.17	0.42	0.52	0.60	0.28	0.66	0.71	0.30	0.37	0.09	0.04	0.39	0.58	0.75	1.00

Goals:	SDG 1	SDG 2	SDG 3	SDG 4	SDG 5	SDG 7	SDG 8	SDG 9	SDG 10	SDG 11	SDG 12	SDG 13	SDG 15	SDG 16	SDG 17	F
SDG1	1.00	-0.02	0.06	0.31	0.08	0.08	0.38	0.40	0.12	0.00	-0.23	0.01	-0.03	0.29	0.28	0.36
SDG2	-0.02	1.00	-0.22	-0.30	-0.05	0.05	-0.07	-0.07	-0.11	-0.14	-0.10	0.34	0.31	-0.18	0.11	-0.11
SDG3	0.06	-0.22	1.00	0.19	0.19	0.08	0.23	0.25	0.46	0.45	0.33	-0.04	0.11	0.17	0.19	0.32
SDG4	0.31	-0.30	0.19	1.00	0.28	0.29	0.45	0.33	0.27	0.26	-0.07	-0.25	0.01	0.45	0.23	0.40
SDG5	0.08	-0.05	0.19	0.28	1.00	0.08	0.23	0.42	0.23	0.15	0.13	0.14	0.34	0.26	0.30	0.41
SDG7	0.08	0.05	0.08	0.29	0.08	1.00	0.19	0.01	0.02	0.04	-0.18	-0.17	0.26	0.20	0.29	0.18
SDG8	0.38	-0.07	0.23	0.45	0.23	0.19	1.00	0.62	0.39	0.31	-0.09	-0.05	0.05	0.57	0.49	0.47
SDG9	0.40	-0.07	0.25	0.33	0.42	0.01	0.62	1.00	0.47	0.38	0.15	0.19	0.15	0.46	0.44	0.61
SDG10	0.12	-0.11	0.46	0.27	0.23	0.02	0.39	0.47	1.00	0.59	0.26	0.00	0.00	0.34	0.19	0.37
SGG11	0.00	-0.14	0.45	0.26	0.15	0.04	0.31	0.38	0.59	1.00	0.26	-0.09	0.03	0.25	0.24	0.34
SDG12	-0.23	-0.10	0.33	-0.07	0.13	-0.18	-0.09	0.15	0.26	0.26	1.00	0.19	-0.03	-0.12	-0.11	0.06
SDG13	0.01	0.34	-0.04	-0.25	0.14	-0.17	-0.05	0.19	0.00	-0.09	0.19	1.00	0.28	-0.08	-0.04	0.00
SDG15	-0.03	0.31	0.11	0.01	0.34	0.26	0.05	0.15	0.00	0.03	-0.03	0.28	1.00	-0.06	0.30	0.28
SDG16	0.29	-0.18	0.17	0.45	0.26	0.20	0.57	0.46	0.34	0.25	-0.12	-0.08	-0.06	1.00	0.30	0.36
SDG17	0.28	0.11	0.19	0.23	0.30	0.29	0.49	0.44	0.19	0.24	-0.11	-0.04	0.30	0.30	1.00	0.60
F	0.36	-0.11	0.32	0.40	0.41	0.18	0.47	0.61	0.37	0.34	0.06	0.00	0.28	0.36	0.60	1.00

Table 3. Kendall  $\tau$  correlation coefficient in the scope of sustainable finance and the selected goals of Agenda 2030 in 2016

Table 4. The selected summar	y statistics calculate	d for taxonomic 1	measures of deve	elopment

Summary statistics	SDG1	SDG2	SDG3	SDG4	SDG5	SDG7	SDG8	SDG9	SDG10	SDG11	SDG12	SDG13	SDG15	SDG16	SDG17	F
Mean	0.49	0.51	0.60	0.53	0.43	0.36	0.53	0.41	0.22	0.49	0.48	0.47	0.47	0.51	0.28	0.40
Median	0.53	0.47	0.63	0.61	0.39	0.31	0.46	0.35	0.19	0.48	0.46	0.53	0.49	0.46	0.18	0.37
Standard deviation	0.25	0.27	0.24	0.26	0.24	0.24	0.24	0.27	0.22	0.25	0.21	0.25	0.28	0.24	0.24	0.25
Coefficient of variation [%]	50.86	53.18	40.99	48.74	56.77	66.36	45.99	64.85	99.59	51.44	43.09	53.65	59.30	48.01	86.59	61.82
Asymmetry	0.21	-0.04	-0.84	-0.49	0.27	0.84	-0.12	0.41	2.15	0.22	0.25	-0.09	0.05	0.21	1.49	0.62

The Netherlands, which took first place due to sustainable finance, also occupied quite good positions in achieving the goals of sustainable development, for eleven of them the position of this country exceeded the average value in the entire group of examined objects.

The positions of individual states in the ranking of countries due to sustainable finances and the implementation of individual goals were often divergent. An example could be Spain, which takes first place, for example, due to goals 3 and 21 due to sustainable finance.

The differences in the classification results obtained between individual goals and area of sustainable finances are also confirmed by the received assessments of Pearson correlation coefficients (Table 2) and Kendall  $\tau$  (Table 3). The highest ratings of correlation coefficients (Person and Kendall  $\tau$ ) were obtained in the case of relationships between sustainable finances and Goals 17 and 9. However, the existence of at least average relationship (0.5 and more) can be confirmed by balanced finances and the goals: 1, 3, 4, 5, 8 and 16. Only in the case of goals: 2, 12 and 13 no relationship was found.

These differences confirm the values of the basic descriptive characteristics of estimated taxonomic measures of development. Coefficients of variation are very high and range from 40% to almost 100% (Table 4). The asymmetry of synthetic measures was also high for sus-

tainable finances and SDG3, SDG7, SDG10 and SDG17 and moderate for goals: SDG4 and SDG9. In the case of this first synthetic measure (for F), the asymmetry is right-sided, which means that for most countries the obtained values of the synthetic measure are lower than the average value. Positive asymmetry also occurs in the case of the taxonomic measure for Goals: 1, 5, 7, 9, 10, 11, 12, 15, 17. The situation is different when analysing the synthetic measure related to the implementation of Goals 3 and 4. In this case, for most countries, the obtained values of the synthetic measure were higher than the average value.

In the next table (Table 5) the results of correspondence analysis were presented. In order to check to what extent eigenvalues of lower dimension spaces describe total inertia, the Greenacre criterion was used (Greenacre, 1994, pp. 141–161; Stanimir, 2005). According to this criterion, eigenvalues that are greater than 1/Q are considered as significant for the study. Since the number of variables in the study is 16, eigenvalues higher than 0.0625 are treated as significant. Table 5 shows that eigenvalues fulfilling this condition are assigned to  $k \le 17$ , while in 17-dimensional space the degree of explanation of total inertia is 94.03%.

The modification proposed by Greenacre significantly improved the quality of mapping. To confirm which space should be applied to the graphic presentation of the co-existence of the categories of variables, the "elbow" criterion was used. This criterion allows finding the place in which the values of modified eigenvalues started to form so-called talus, which means that the differences between the values are getting smaller. The talus started forming from the fourth eigenvalues. It means that this dimension will be appropriate (Figure 1).

k	Eigen- values $\gamma_k$	Singular values $\lambda_k$	Percentage of Inertia $\lambda_k / \lambda$	Cumulative Percentage $\tau_k$	Eigen- values $\tilde{\lambda}_k$	Percentage of Inertia $\tilde{\lambda}_k / \tilde{\lambda}$	Cumulative Percentage $\tilde{\tau}_k$
1	0.647	0.419	13.955	13.955	0.389	16.658	16.658
2	0.524	0.275	9.165	23.120	0.243	10.399	27.057
3	0.519	0.269	8.979	32.099	0.237	10.160	37.217
4	0.480	0.230	7.674	39.773	0.198	8.490	45.707
5	0.468	0.219	7.285	47.059	0.187	7.997	53.704
6	0.448	0.201	6.695	53.754	0.169	7.251	60.955
7	0.424	0.180	6.003	59.756	0.149	6.384	67.339
8	0.387	0.150	5.000	64.756	0.120	5.143	72.482
9	0.367	0.134	4.478	69.234	0.105	4.506	76.988
10	0.351	0.124	4.118	73.352	0.095	4.071	81.059
11	0.335	0.112	3.740	77.092	0.084	3.619	84.678
12	0.324	0.105	3.489	80.581	0.078	3.321	87.999
13	0.318	0.101	3.366	83.947	0.074	3.177	91.177
14	0.303	0.092	3.062	87.009	0.066	2.821	93.998
15	0.273	0.074	2.483	89.492	0.050	2.159	96.156
16	0.265	0.070	2.341	91.833	0.047	1.999	98.156
17	0.257	0.066	2.202	94.034	0.043	1.844	100.000
					$\tilde{\lambda}_k = 2.334$		

Table 5. The results of original and modified versions of correspondence analysis

To present unequivocal interpretation, the Ward method was applied to group the categories of examined variables on the basis of modified values of eight coordinates determining their location in eight-dimensional space. The results of relationships between the categories of the examined variables using the Ward method are presented in Figure 2. The critical value of the distance at which the combination of classes (4.15) was determined according the measure proposed by Grabiński (2003). A critical value is marked in Figure 2 with a horizontal line; the typological groups are also indicated.

The results of the study confirm the relationship between the sustainable finance model and the implementation of SDGs. The first typological group (the highest-ranked positions



Figure 1. Eigenvalue chart (source: own elaboration based on Table 5)



Figure 2. The results of the correspondence analysis (source: own elaboration)

according to SDGs and sustainable finance in the rankings) includes the Scandinavian countries (Denmark, Finland, and Sweden). The Netherlands also ranked highly referring to sustainable finance and SDGs. The lowest ranking positions according to sustainable finance were taken by Hungary, Lithuania, and Spain. Spain's case is interesting because, compared with Hungary and Lithuania, Spain has better achieved SDG, SDG13 and SDG15, but due to financing and low share of revenues from environmental taxes and social contribution, this country is classified in the lowest typological group of sustainable finance. The same indicators explain the position of France in sustainable finance ranking, although France is above average in achievement of SDG9, SDG10, SDG12 and SDG13 compared with the other countries in the same typological group. None of the countries simultaneously ranked first in the implementation of the SDGs (1, 2, 3, 4, 5, 7, 8, 9, 10, 11, 12, 13, 15, 16, 17) and the highest place in the sustainable finance. None of the analysed countries simultaneously had the highest level of SDGs implementation for all SDGs. Sweden, Denmark, and Finland were closest to achieving all SDGs in the top ranking positions, but they did not manage to achieve the top ranking position (1) but were over average (2) for SDG3 and SDG10. All Scandinavian countries and the Netherlands were ranked in the top in sustainable finance. The Scandinavian countries and the Netherlands represent the sustainable finance model 3.0, which is the highest level of sustainable finance. Typological groups of countries according to sustainable finance model and SDGs are presented in Table 6.

Countries applying the sustainable finance model 3.0 were the only group fully implementing all of the SDGs analysed, in particular the SDG on climate action. This group of countries is distinguished by a developed system of environmental taxes and the lowest level of income inequality measured by the GINI coefficient. It is also a group of countries with the highest level of sustainability awareness. The sustainable finance model 3.0 is based on public and private financing that are coherent and integrated with each other; the proportions between public and private financing are balanced. The financial market is well developed and many different kinds of sustainable financial instruments are offered by financial institutions.

Countries applying sustainable finance model 2.0 also have a well-developed environmental tax system, but in this group of countries, some countries produce the largest source of GHG emissions in Europe (Germany, Luxemburg, and the U.K.); hence, these countries

Conventional finance F*	Sustainable finance model 1.0. F > S, E	Sustainable finance model 2.0. T = F + S + E	Sustainable finance model 3.0. S, E > F
Hungary, Lithuania, Spain	Czech Republic, France, Greece, Ireland, Italy, Latvia, Poland, Portugal, Slovakia	Austria, Belgium, Estonia, Germany, Luxemburg, Slovenia, United Kingdom	Denmark, Finland, the Netherlands, Sweden
SDG1, SDG2	SDG1, SDG10, SDG12	SDG1, SDG3, SDG4, SDG9, SDG10, SDG11, SDG16	SDG1, SDG2, SDG3, SDG4, SDG5, SDG7, SDG8, SDG9, SDG10, SDG13, SDG15, SDG16, SDG17

Table 6. Relationship between sustainable finance model and SDGs

*Note:* \*F, S, E-dimensions: financial, environmental, social; T – integrated, total value (Schoenmaker, 2017).

have not effectively implemented SDG7, SDG13, SDG17 related to climate change. Public and private financing in model 2.0 are coherent, integrated, and balanced. The role of the financial sector and public financial sector markets in financing of sustainable development are interdependent.

The sustainable finance model 1.0, or conventional finance model, is applied in countries after systemic transformation and states that were most affected by the effects of the 2008 financial crisis. The economies of these countries applying the 1.0 sustainable finance model are based on fossil fuels (coal) and are only starting to adjust activities to replace fossil fuels with renewable energy sources. The environmental taxes system in this group of countries is developing, and these countries focus their efforts on poverty with social programs. The conventional model and model 1.0 are mainly based on public sector financing and the financial market is not adjusted to meet the expectations of sustainable development financing.

The research results show that the implementation of SDGs depends on applying a sustainable finance model. The more sustainable the finance model, the more the SDGs are implemented, with the most crucial impact of sustainable finance model focusing on the environmental pillar of sustainable development (climate action). The less sustainable the finance model, the more visible the focus on socially-related goals.

Comparing the study results with the sustainability rankings, the following conclusions are noteworthy:

- according to the Sustainable Competitiveness Index 2019 (SolAbility, 2019) the top 5 ranking positions are occupied by Scandinavia: Sweden is leading the Sustainable Competitiveness Index – followed by the other the Scandinavian nations;
- in 2017 all Scandinavian countries (Denmark, Sweden Finland, Norway) were on the top 5 ranking position of the Country Sustainability Ranking which analysis countries' ESG performance;
- Sweden is a leader in sustainable financing; its financial institutions have been some of the first active in the green bond market (SEB, Nordea, Kommuninvest, Vasakronan) (Stockhol Sustainable Finance Centre);
- The Nordic Model of governance and public service provision is all about sustainability and social cohesion (The Green Bond market in the Nordics 2018 (Climate Bonds Initiative, 2018));
- Sweden is the sixth largest source of labelled green bond issuance; Norway, Denmark and Finland are in the Top 20 (Climate Bonds Initiative, 2018);
- SDG index for the European Union (2019) revels Denmark, Sweden, Finland are on 3 top ranking positions regarding to achievement of SDGs (Sustainable Development Solutions Network and Institute for European Environmental Policy, 2019).

The analysis of sustainable rankings results confirm that the study results are coherent are in line with the rankings. The SDG index is especially important as it is in line with our research results not only for the top on the list but also for other (middle and last) ranking positions. The Report points out what we found in our study that the greatest challenge is to secure achieve of SDGs related to climate change and climate actions among analysed country group.

Based on thy analysis of related work we draw the conclusion that there are no research results we can compare and discuss directly with our study results. There is a research gap in the scope the study we have discuss. No study has been dealing with the similar to ours research hypothesis: the more sustainable the finance model, the better the link with achievement of SDGs.

Our research results are in consonance with the results of Gambetta et al. (2019). The authors analysed the design of sustainable finance instruments for financing SDGs, including, among others, environmental taxes and public expenditures for social and environmental goals. The Uruguay sustainable finance model (model 1.0.) is in the early stage of development: SDG1 has been achieved totally whereas others SDGs remain a challenge. Financing sustainable development plays a leading role in the public sector. The authors concluded that there is a need for an in-depth study of finance and sustainability.

Our results also align with Discover the SDGs implementation level a year and a half after ratification of 2030 Agenda (Discover the SDGs, 2015; Sustainability for all, 2019). According to the report, Sweden (84.5), Denmark (83.9), Norway (82.3), and Finland (81.0) are in the leading positions, mainly due to their good performance in social and economic issues, although the data showed that they must still work on the transition to a low carbon economy. Spain ranked 30th with 72.2; United Kingdom needed to improve in Zero hunger and Affordable and clean energy (Discover the SDGs, 2015; Sustainability for all, 2019).

The results of the analysis are difficult to compare given the background of related work because such research has not yet been conducted. This is one of the first approaches to examine the relationship between sustainable finance model and SDGs. Much attention has been paid to the role of public expenditure or environmental taxes in achieving SDGs, whereas focus on the relationships with the sustainable finance model is lacking.

#### Conclusion and recommendations

The implementation of SDGs in the face of climate change, social exclusion, and increasing inequalities is a priority for governments around the world. Simultaneously, implementation of SDGs is not possible without providing funds for their financing. Discussion of the design of sustainable financial systems and sustainable finance model is underway and research on the effectiveness of financing for sustainable development is in an early stage. This paper is one of the first based on an original research approach analysing the relationship between sustainable finance model and SDGs using correspondence analysis. Such analyses have not been conducted before to analyse the link between SDGs and sustainable finance model, which indicates the novelty of our research approach.

The findings verify the main hypothesis of the link between a sustainable finance model and SDGs, indicating that that the more sustainable the finance model, the better the results of a given country in achieving SDGs. The study was conducted for fifteen SDGs (SDGs 1,2, 3,4,5,7,8,9,10,11,12,13,15,16,17). In case of SDG6 and SDG14 the variables are not available. We analysed 23 EU countries belonging to the OECD for 2016. The analysis was based on variables from the OECD database; a total of 98 variables were included in the analysis, of which 83 represented SDGs and 15 represented sustainable finance. As a result of the study, it was shown that SDG1 has been achieved in all sustainable finance models. This is justified by the fact that the sustainable public financial system in its basic form, through the redistributive function of public finances, guarantees the equalization of income and through social policy provides access for all citizens to basic public goods and services. As part of these basic public goods and services, there is the fight against social exclusion. For the same reason, SDG4, SDG10 have been related to all sustainable finance models. SDG3, SDG4, SDG5, SDG7, SDG8, SDG9, SDG13 and SDG16 due to the fact that they concern and have an impact on well being and quality of life are not possible to achieve only on the basis of a sustainable public financial system. They require also cooperation and involvement of the sustainable market financial system. This is due to, among others from the fact that health and wellbeing is influenced by, among others air quality that can be achieved by implementing technological innovations primarily in the business sector and the state is not able to provide sufficient public funding in this regard, nor in the field of financing innovation. Integrated actions are required here through fiscal policy (public expenditure, taxes) and dedicated sustainable financial products and services and procedures (sustainable benchmarks, sustainable ratings, ESG risk management).

The study results confirm the relationship between the sustainable finance model and implementation of SDGs, which was strongest in countries applying sustainable finance model 3.0. In these countries, both the public and private financial systems are involved in financing sustainable development; these systems interpenetrate and complement each other.

The sustainable finance model 2.0 allows for above average SDGs implementation; however, financing environmental goals requires adaptation to the needs and structure of economies. Countries in this group are characterized by high greenhouse gas emissions. Therefore, processes of adaptation of both public policies and the financial market are required to reduce emissions using environmental taxes and financial products to finance technologies supporting the development of renewable energy sources.

The other two models of conventional finance and sustainable finance model 1.0 included a group of countries whose both public and private financial markets were in the phase of adaptation to the needs of financing sustainable development. The public sector plays a major role in financing sustainable development. Countries from this group achieved SDG1 without any problems, whereas the implementation of the other SDGs was problematic.

This research is pioneering: the relationship between sustainable finance model and SDGs has not been previously studied. Hence, we encountered numerous difficulties during the study development and literature review stages, as well as given the lack of data. The results of the study could contribute to further in-depth research for a broader time perspective and a larger number of variables. We plan to develop analyses based on fuzzy cognitive methods based on data from in-depth interviews, complementary to statistical data that are incomplete and difficult to compare over time for individual countries. Based on the results of the study, the following recommendations were formulated:

- Effective achievement of SDGs requires an effective and integrated financial model based on the public and market financial systems, which are interdependent and related in terms of meeting the criteria of sustainable financing. This is why governments should ensure parallel development and cooperation between public and market financial systems toward sustainability;
- The public financial system upon which the conventional model and the 1.0 model are primarily based is able to provide support for the implementation of the social pillar of sustainable development, but is not able ensure the achievement of environmental

goals; therefore, governments should provide support from financial institutions for the public financial system for financing environmental goals through legal regulations;

- Due to the visible impact of financing on SDG, governments should introduce effective micro- and macro-prudential solutions to control this phenomenon in the financial system;
- Integration of financial policy and sustainable development policy should be a priority for governments; these policies should have common monitoring indicators and common goals to ensure integration of the financial sphere and reality;
- Governments should introduce a system for reporting and monitoring the impact of the public and market financial system on SDG achievement.

## Acknowledgements

Research results presented in this paper are an element of research project implemented by the National Science Centre Poland under the grant OPUS13 no 2017/25/B/HS4/02172.

## Funding

The research is financed by National Science Centre Poland (Narodowe Centrum Nauki), grant no OPUS13 no 2017/25/B/HS4/02172.

## Author contributions

Conceptualization: M.Z.; methodology: I.B., K.Ch.; software: I.B., K.Ch.; validation: I.B., K.Ch.; formal analysis M.Z., I.B., K.Ch.; investigation: M.Z.; resources: MZ. I.B., K.Ch.; data curation: I.B., K.Ch.; writing original draft preparation: M.Z. I.B., K.Ch.; writing review and editing: M.Z., I.B., K.Ch.; visualization: M.Z.; I.B., K.Ch.; supervision: M.Z.; project administration: M.Z.; funding acquisition: M.Z.

## **Disclosure statement**

Authors don't have any competing financial, professional, or personal interests from other parties.

## References

- Alm, K., & Sievänen, R. (2013). Institutional investors, climate change and human rights. *Journal of Sustainable Finance & Investment*, 3(3), 177–183. https://doi.org/10.1080/20430795.2013.791139
- Andersen, E. B. (1991). *The statistical analysis of categorial data*. Springer-Verlang. https://doi.org/10.1007/978-3-642-97353-6
- Apergis, N., Eleftheriou, S., & Payne, J. E. (2013). The relationship between international financial reporting standards, carbon emissions, and R&D expenditures: Evidence from European manufacturing firms. *Ecological Economics*, 88, 57–66. https://doi.org/10.1016/j.ecolecon.2012.12.024

- Aspinall, N. G., Jones, S. R., Mc Neill, E. H., Werner, R. A., & Zalk, T. (2018). Sustainability and the financial system. Review of literature 2015. *British Actuarial Journal*, 23, e10. https://doi.org/10.1017/S1357321718000028
- Bąk, I. (2013). Statystyczna analiza aktywności turystycznej seniorów w Polsce. Wydawnictwo Uczelniane ZUT w Szczecinie.
- Bąk, I., Szczecińska, B., & Cheba, K. (2019). The impact of transport on the quality of the environment in cities in Poland – a statistical analysis. *Transport Research Procedia*, 39, 24–33. https://doi.org/10.1016/j.trpro.2019.06.004
- Barroy, H., Kutzin, J., Tandon, A., Kurowski, Ch., Lie, G., Borowitz, M., Sparkes, S., & Dale, S. (2018). Assessing fiscal spacefor health in the SDG era: A different story. *Health Systems & Reform*, 4(1), 4–7. https://doi.org/10.1080/23288604.2017.1395503
- Burchardt, T., & Vizard, P. (2007). Definition of equality and framework for measurement: Final Recommendations of the Equalities Review Steering Group on Measurement. CASE paper 120. LSE.
- Buse, K., & Hawkes, S. (2015). Health in the sustainable development goals: Ready for a paradigm shift? Global Health, 11, 13. https://doi.org/10.1186/s12992-015-0098-8
- Centemeri, L. (2009). Environmental damage as negative externality: Uncertainty, moral complexity and the limits of the market. https://doi.org/10.4000/eces.266
- Chen, S. (2013). Green finance and development of low carbon economy. In F. Chen, Y. Liu, & G. Hua (Eds.), LTLGB 2012. Proceedings of International Conference on Low-carbon Transportation and Logistics, and Green Buildings. Springer. https://doi.org/10.1007/978-3-642-34651-4
- Čihák, M., Demirgüç-Kunt, A., Feyen, E., & Levine, R. (2012). Benchmarking financial systems around the world (Policy Research Working paper 6175, pp. 1–56). https://doi.org/10.1596/1813-9450-6175
- Clausen, S. E. (1998). Applied correspondence analysis: An introduction. Sage Publications.
- Climate Bonds Initiative. (2018). The Green Bond market in the Nordics 2018. https://www.google. com/url?sa=t&rct=j&q=&esrc=s&source=web&cd=&ved=2ahUKEwjm7azb\_\_\_\_sAhUDxYsKHceb CeYQFjAAegQIAxAC&url=https%3A%2F%2Fwww.climatebonds.net%2Ffiles%2Freports%2Fcbinordics-final-03b.pdf&usg=AOvVaw0BUeY1pOUhGrV9XtAEaMbY
- Collste, D., Pedercini, M., & Cornell, S. E. (2017). Policy coherence to achieve the SDGs: Using integrated simulation models to assess effective policies. *Sustainability Science*, *12*(6), 921–931. https://doi.org/10.1007/s11625-017-0457-x
- Development Finance International, & Oxfam. (2015). Financing the sustainable development goals: Lessons from government spending on the MDGs (report). Government Spending Watch. Retrieved June 5, 2020, from www.governmentspendingwatch.org
- Discover the SDGs. (2015). https://www.sdgindex.org/reports/sustainable-development-report-2019/
- Edmans, A. (2011). Does the stock market fully value intangibles? Employee satisfaction and equity prices. *Journal of Financial Econ*omics, 101(3), 621–640.
- Elder, M., & Olsen, H. O. (2019). The design of environmental priorities in the SDGs. *Global Policy*, 10(S1) Special Issue: Knowledge and Politics in Setting and Measuring SDGs, 70–82. https://doi.org/10.1111/1758-5899.12596
- European Communities. (2002). European social statistics. Income, poverty and social exclusion (2<sup>nd</sup> report). https://ec.europa.eu/eurostat/ramon/coded\_files/KS-BP-02-008-EN.pdf
- FAO. (2020). Retrieved July 27, 2020, from http://www.fao.org/3/y4137e/y4137e02b.htm
- Ferreira, M. C., de Carvalho, R., Amorim, S. V., Kimura, H., & de Moraes Barboza, F. L. (2016). A systematic review of literature about finance and sustainability. *Journal of Sustainable Finance & Investment*, 6(2), 112–147. https://doi.org/10.1080/20430795.2016.1177438

- Fullwiler, S. T. (2015). *Sustainable finance: Building a more general theory of finance* (Working Paper No. 106). Binzagr Institute for Sustainable Prosperity.
- Gain, A. K., Giupponi, S., & Wada, Y. (2016). Measuring global water security towards sustainable development goals. *Environmental Research Letters*, 11, 124015. https://doi.org/10.1088/1748-9326/11/12/124015
- Gambetta, N., Azadian, P., Hourade, V., & Reyes, M. E. (2019). The financing framework for sustainable development in emerging economies: The case of Uruguay. *Sustainability*, 11(4), 1059. https://doi.org/10.3390/su11041059
- Gerster, R. (2011). Sustainable finance: Achievements, Challenges, Outlook. Gerster Consulting, Richterswill.
- Gompers, P., Ishii, J., & Metrick, A. (2003). Corporate Governance and equity prices. *Quarterly Journal* of Economics, 118(1), 107–156. https://doi.org/10.1162/00335530360535162
- Goodman, L. A. (1986). Some useful extensions of the usual correspondence analysis approach and usual log-linear models approach in the analysis of contingency tables. *International Statistical Review*, 54(3), 243–270. https://doi.org/10.2307/1403053
- Grabiński, T. (2003). *Taxonomic analysis from regional perspective* (translated from Polish). Wydawnictwo Akademii Ekonomicznej w Krakowie.
- Greenacre, M. (1984). Theory and applications of correspondence analysis. Academic Press.
- Greenacre, M. (1993). Correspondence analysis in practice. Academic Press.
- Greenacre, M. (1994). Multiple and joint correspondence analysis. In M. Greenacre & J. Blasius (Eds.), Correspondence analysis in social sciences. Recent developments and applications (pp. 141–161). Academic Press.
- Greenacre, M., & Hastie, T. (1987). The geometric interpretation of correspondence analysis. *Journal of the American Statistical Association*, 82(398), 437–447. https://doi.org/10.1080/01621459.1987.10478446
- Gupta, J., & Vegelin, C. (2016). Sustainable development goals and inclusive development. *International Environmental Agreements: Politics, Law and Economics, 16,* 433–448. https://doi.org/10.1007/s10784-016-9323-z
- Hajer, M., Nilsson, M., Raworth, K., Bakker, P., Berkhout, F., De Boer, Y., Rockström, J., Ludwig, K., & Kok, M. (2015). Beyond Cockpit-ism: four insights to enhance the transformative potential of the sustainable development goals. *Sustainability*, 7(2), 1651–1660. https://doi.org/10.3390/su7021651
- Hebb, T. (2013). Impact investing and responsible investing: what does it mean? *Journal of Sustainable Finance & Investment*, 3(2), 71–74. https://doi.org/10.1080/20430795.2013.776255
- Heeks, R., Mirta, A., Nishant, S., & Kintu, R. (2013). Inclusive innovation: Definition, conceptualisation and future research priorities (Development Informatics Working Paper no. 53). https://doi.org/10.2139/ssrn.3438439
- International Monetary Fund. (2015). From billions to trillions transforming development finance, post-2015 financing for development: Multilateral development finance. DC2015-0002. https://www.imf.org/en/News/Articles/2015/09/14/01/49/pr15170
- International Monetary Fund (2019). *Fiscal policies for Paris climate strategies: From principle to practice* (IMF Policy Paper). Washington, DC.
- Jeucken, M. H. A. (2004). Sustainability in finance A retroductive exploration [PhD thesis. Erasmus University Rotterdam]. Eburon, Delft.
- Kahn, M., Mohaddes, K., Ng, R., Pesaran, H., Raissi, M., & Yang, J. Ch. (2019). Long-term macroeconomic effects of climate change: A cross-country analysis (Working Paper). Federal Reserve Bank of Dallas, Globalization Institute.

- Kedir, A., Elhiraika, A., Chinzara, Z., & Sandjong, D. (2017). Growth and development finance required for achieving SDGs in Africa. *African Development Review*, 29(1), 15–26. https://doi.org/10.1111/1467-8268.12230
- Kharas, H., Prizzon, A., & Rogerson, A. (2015). Financing the post-2015 sustainable development goals. A rought map. Overseas Development Institute, London. https://www.odi.org/sites/odi.org.uk/files/ odi-assets/publications-opinion-files/9373.pdf
- Klees, S. J. (2017). Financing education and all the other SDGs: Global taxation is needed. In *Education*, training and Agenda 2030: What progress one year on? NORRAG NEWS 54. https://www.norrag.org/ fileadmin/Full%20Versions/NN54.pdf#page =118
- Kumar, S., Kumar, N., & Vivekadhish, S. (2016). Millennium Development Goals (MDGs) to Sustainable Development Goals (SDGs): Addressing unfinished agenda and strengthening sustainable development and partnership. *Indian Journal of Community Medicine*, 41(1), 1–4. https://doi.org/10.4103/0970-0218.170955
- Le Blanc, D. (2015). Towards integration at last? The Sustainable Development Goals as a network of targets. Sustainable Development, 23(3), 176–187. https://doi.org/10.1002/sd.1582
- Lebart, L., Morineau, A., & Warwick, K. M. (1984). Multivariate descriptive statistical analysis. Correspondence analysis and related techniques for large matrices. John Wiley & Sons.
- Lee, K. H., & Min, B. (2015). Green R&D for eco-innovation and its Impact on carbon emissions and firm performance. *Journal of Cleaner Production*, 108(Part A), 534–542. https://doi.org/10.1016/j.jclepro.2015.05.114
- Lin, B., & Li, X. (2011). The effect of carbon tax on per capita CO<sub>2</sub> emissions. *Energy Policy*, 39(9), 5137–5146. https://doi.org/10.1016/j.enpol.2011.05.050
- Long, C., & Miller, M. (2017). Taxation and the Sustainable Development Goals. Do good things come to those who tax more? Shaping policy for development. ODE Briefing note.
- Nerini, F., Tomei, J., & To, L. S. (2018). Mapping synergies and trade-offs between energy and the Sustainable Development Goals. *Nature Energy*, 3, 10–15. https://doi.org/10.1038/s41560-017-0036-5
- Nikolakis, W., Cohen, D., & Nelson, H. (2012). What matters for socially responsible investment (SRI) in the natural resources sectors? SRI mutual funds and forestry in North America. *Journal of Sustainable Finance and Investment*, 2(2), 136–151.
- Nilsson, M., Chisholm, E., Griggs, D., Howden-Chapman, P., McCollum, D., Messerli, P., Neumann, B., Stevance, A. S., Visbeck, M., & Stafford-Smith, M. (2018). Mapping interactions between the sustainable development goals: lessons learned and ways forward. *Sustainability Science*, 13, 1489– 1503. https://doi.org/10.1007/s11625-018-0604-z
- Nilsson, M., Griggs, D., & Visbeck, M. (2016). Map the interactions between Sustainable Development Goals. Nature, 534, 320–322. https://www.nature.com/news/policy-map-the-interactions-betweensustainable-development-goals-1.20075
- Oates, W. E. (1995). Green taxes: Can we protect the environment and improve the tax system at the same time? *Southern Economic Journal*, 61(4), 915–922. https://doi.org/10.2307/1060731
- OECD. (2018). Global outlook on financing for sustainable development 2019. Time to face the challenge. United Nations Environmental Programme Financial Initiative. Retrieved June 4, 2020, from https://www.oecd.org/development/global-outlook-on-financing-for-sustainable-development-2019-9789264307995-en.htm
- Parry, I. W., & Small, K. A. (2005). Does Britain or the United States have the right gasoline tax? American Economic Review, 95(4), 1276–1289. https://doi.org/10.1257/0002828054825510
- Pisano, U., Martinuzzi, A., & Bruckner, B. (2012). *The financial sector and sustainable development:* Logics, principles and actors (ESDN Quarterly Report No 27).
- Principles for Responsible Investment. (2017). The SDG investment case. London.

- Rambla, F. X., & Langthaler, M. (2016). *The SDGs and inclusive education for all from special education to addressing social inequalities*. Austrian Foundation for Development Research.
- Rubin, E. S. (2011). Innovation and Climate Change. In *Innovation. Perspectives for the 21st Century*. BBVA.
- Sachs, J. D. (2015). Goal-based development and the SDGs: Implications for development finance. Oxford Review of Economic Policy, 31(3-4), 268–278. https://doi.org/10.1093/oxrep/grv031
- Schmidt-Traub, G., & Sachs, D. (2015, April). Financing sustainable development: implementing the SDGs through effective investment strategies and partnerships (Working Paper). Sustainable Development Solution Network.
- Schoenmaker, D. (2017). Investing for the common good: A sustainable finance framework. Bruegel essay and lecture series. https://bruegel.org/wp-content/uploads/2017/07/From-traditional-to-sustainable-finance\_ONLINE.pdf
- Scholtens, B. (2006). Finance as a driver of corporate social responsibility. *Journal of Business Ethics*, 68(1), 19–33. https://doi.org/10.1007/s10551-006-9037-1
- SolAbility. (2019). The Sustainable Competitiveness Index 2019. http://solability.com/global-sustainablecompetitiveness-index/the-global-sustainable-competitiveness-index-2019
- Stafford-Smith, M., Griggs, D., Gaffney, O., Ullah, F., Reyers, B., Kanie, N., Stigson, B., Shrivastava, P., Leach, M., & O'Connell, D. (2017). Integration: the key to implementing the Sustainable Development Goals. Sustainability Science, 12, 911–919. https://doi.org/10.1007/s11625-016-0383-3
- Stanimir, A. (2005). *Correspondence analysis as a tool for studying economic phenomena* (translated from Polish). Wydaw. AE we Wrocławiu.
- Sustainability for all. (n.d.). Are countries achieving the Sustainable Development Goals? Discover the SDGs implementation level a year and a half after ratification of 2030 Agenda. https://www.ac-tivesustainability.com/sustainable-development/are-countries-achieving-the-sustainable-development-goals/
- Sustainable Development Solutions Network and Institute for European Environmental Policy. (2019). *Europe Sustainable Development Report*. https://www.sdgindex.org/reports/2019-europe-sustainable-development-report/
- The Sustainable Development Goals Center for Africa. (2017, January 27). SDG financing for Africa: key propositions and areas of engagement. Discussion Paper for Development Finance Workshop. In *The Sustainable Development Goals Center for Africa Conference*, Kigali, Rwanda. https://sdgcafrica.org/wp-content/uploads/2017/03/sdg-financing-for-africa\_key-propositions-andareas-of-engagement-.pdf
- UNEP. (2016). Fiscal policies and the SDGs. Green economy. Policy brief. https://greenfiscalpolicy.org/ policy\_briefs/fiscal-policies-and-the-sdgs-unep-policy-brief/
- United Nations. (2015). Transforming our world: The 2030 Agenda for sustainable development A/RES/70/1. United Nations General Assembly, New York.
- United Nations. (2016). The sustainable development goals. Report. New York.
- United Nations. (2019). More money needed to implement Sustainable Development Goals, Secretary-General tells ECOSOC Financing for Development Forum, calling 2019 "Defining Year". Secretary-General. Statements and Messages. SG/SM/19546-ECOSOC/6973-DEV/3397. Retrieved April 15, 2019, from https://www.un.org/press/en/2019/sgsm19546.doc.htm
- United Nations Global Compact. (2019). Scaling finance for the Sustainable Development Goals. Foreign direct investment, financial intermediation and public-private partnerships. https://sdghub.com/ project/scaling-finance-for-the-sustainable-development-goals-foreign-direct-investment-financialintermediation-and-public-private-partnerships/

- Vandekerckhove, W., & Leys, J. (2012). Dear Sir, We are Not an NGO. *The Journal of Sustainable Finance* & *Investment*, 2(2), 152–161.
- Waring, P., & Edwards, T. (2008). Socially responsible investment: explaining its uneven development and human resource management consequences. *Corporate Governance: An International Review*, 16(3), 135–145. https://doi.org/10.1111/j.1467-8683.2008.00676.x
- Waygood, S. (2011). How do the capital markets undermine sustainable development? What can be done about this? *Journal of Sustainable Finance and Investment*, 1(1), 81–7. https://doi.org/10.3763/jsfi.2010.0008
- World Economic Forum. (2020). *Global risk report*. Retrieved June 4, 2020, from https://www.weforum. org/reports/the-global-risks-report-2020
- Ziolo, M., Filipiak, B. Z., Bąk, I., & Cheba, K. (2019). Finance, sustainability and negative externalities. An overview of the European context. *Sustainability*, 11(15), 4249. https://doi.org/10.3390/su11154249
- Zorlu, P. (2018). Transforming the financial system for delivering sustainable development A high-level overview. Finance Taskforce (IGES Discussion Paper). Institute for Global Environmental Strategies. Retrieved June 3, 2019, from https://www.iges.or.jp/en/pub/transforming-financial-systemdelivering/en